

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Were, Maureen; Mollel, Lekinyi

Working Paper Public debt sustainability and debt dynamics: The case of Tanzania

WIDER Working Paper, No. 2020/112

Provided in Cooperation with: United Nations University (UNU), World Institute for Development Economics Research (WIDER)

Suggested Citation: Were, Maureen; Mollel, Lekinyi (2020) : Public debt sustainability and debt dynamics: The case of Tanzania, WIDER Working Paper, No. 2020/112, ISBN 978-92-9256-869-6, The United Nations University World Institute for Development Economics Research (UNU-WIDER), Helsinki,

https://doi.org/10.35188/UNU-WIDER/2020/869-6

This Version is available at: https://hdl.handle.net/10419/229336

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



WIDER Working Paper 2020/112

Public debt sustainability and debt dynamics

The case of Tanzania

Maureen Were¹ and Lekinyi Mollel²

September 2020

In partnership with



United Nations University World Institute for Development Economics Research

wider.unu.edu

This publication results from Sustainable development solutions for Tanzania – strengthening research to achieve SDGs, a collaborative project between the UONGOZI Institute in Dar es Salaam, Tanzania, and UNU-WIDER in Helsinki, Finland, with a main research objective of informing the development and implementation of policies aiming for economic transformation and sustainable development in Tanzania and the East African region. With financial support provided by the Ministry for Foreign Affairs of Finland, the joint project launched in 2018 with key questions giving the partners a framework for collaboration and the research work to be undertaken. The project focuses on macroeconomic perspectives, domestic resource mobilization, extractives, industrialization, sustainable livelihoods, and gender as a cross-cutting issue. The project provides local stakeholders a platform for research and policy discussions on Tanzania and bridges these discussions to the regional and international development debate.

About UONGOZI Institute

'Uongozi' means leadership in Kiswahili, and inspiring and strengthening leadership is the core purpose of our organisation. UONGOZI Institute is dedicated to supporting African leaders to attain sustainable development for their nations and for Africa. This is done through the provision of high-quality executive education (leadership competencies), facilitation of policy dialogues, action-oriented research and technical assistance for public and private institutions.

UONGOZI Institute, Magogoni Street, Kivukoni, P.O. Box 105753, Dar es Salaam, Tanzania

Abstract: Rising public debt in sub-Saharan Africa remains a matter of concern. We provide an analysis of public debt and debt sustainability in Tanzania, focusing on external debt. Though current and previous analyses using the IMF-World Bank debt sustainability framework indicate low risk of public external debt distress, these analyses are sensitive to exchange rate volatility and export shocks and are predicated on strong assumptions of robust future economic growth and reduced government borrowing. Moreover, empirical evidence of debt sustainability based on the fiscal reaction function approach is weak. The challenge lies in ensuring debt remains sustainable, given the need to scale up development expenditure to address infrastructure gaps amid dwindling donor financing and vulnerability to exogenous shocks, particularly in light of the COVID-19 pandemic. Rapid debt accumulation—particularly commercial debt—could expose Tanzania to external risks. Leveraging on concessional borrowing, efficient public investment, enhanced debt management, and domestic resource mobilization are critical.

Key words: public debt, Tanzania, debt sustainability, external debt

JEL classification: E60, E62, H63, H68

Acknowledgements: The authors would like to thank Benno Ndulu and Nicas Yabu for their valuable comments on earlier versions of the paper. Any errors and omission are ours.

¹ UNU-WIDER, Helsinki, Finland, corresponding author: were@wider.unu.edu; ² Bank of Tanzania, Dar es Salaam, Tanzania.

This study has been prepared within the project Sustainable development solutions for Tanzania – strengthening research to achieve SDGs funded by the Ministry for Foreign Affairs of Finland.

Copyright © UNU-WIDER 2020

Information and requests: publications@wider.unu.edu

ISSN 1798-7237 ISBN 978-92-9256-869-6

https://doi.org/10.35188/UNU-WIDER/2020/869-6

Typescript prepared by [xxx].

United Nations University World Institute for Development Economics Research provides economic analysis and policy advice with the aim of promoting sustainable and equitable development. The Institute began operations in 1985 in Helsinki, Finland, as the first research and training centre of the United Nations University. Today it is a unique blend of think tank, research institute, and UN agency—providing a range of services from policy advice to governments as well as freely available original research.

The Institute is funded through income from an endowment fund with additional contributions to its work programme from Finland, Sweden, and the United Kingdom as well as earmarked contributions for specific projects from a variety of donors.

Katajanokanlaituri 6 B, 00160 Helsinki, Finland

The views expressed in this paper are those of the author(s), and do not necessarily reflect the views of the Institute or the United Nations University, nor the programme/project donors.

1 Introduction

The rate at which Africa's public debt has accumulated continues to attract attention regionally and internationally. According to IMF (2018a), the median level of public debt in sub-Saharan Africa (SSA) as at end of 2017 exceeded 50 per cent of gross domestic product (GDP).¹ About 40 per cent of low-income developing countries in SSA had slid into debt distress or were at high risk of debt distress.² The average level of SSA debt was about 57 per cent of GDP in 2019.

Tanzania's public debt, though rising, has not attracted much attention compared with that of other countries in the region, such as Ethiopia and Kenya. The country has generally been considered to be at low risk of debt distress (IMF 2018b). Nonetheless, the rate of debt accumulation coupled with the increase in public investment needs, including the funding of strategic infrastructure projects in the context of a relatively low tax base and dwindling donor funding, pose a challenge. In addition, it has been noted that low debt-to-GDP ratios for Tanzania are partly due to debt relief provided in the early 2000s under the frameworks of the Heavily Indebted Poor Countries (HIPC) Initiative and the Multilateral Debt Relief Initiative (MDRI). Moreover, the changing composition of debt from concessional towards the relatively more expensive commercial debt pose a challenge of sustainability in future. Not only that, but also access to international financial markets has introduced new risks, such as vulnerability to changes in financing conditions.

Though the debt-to-GDP ratio for Tanzania seems reasonable-37.8 per cent as of June 2018the debate about the (in)adequacy of the debt sustainability framework of the International Monetary Fund (IMF) and World Bank that is commonly used in the assessment remains a challenge. Part of the concern relates to the fact that the focus on the level of public debt as measured by debt-to-GDP ratio provides an incomplete picture and hence should not be the only or main yardstick for determining sustainability. For instance, the ratio changes when GDP level rises due to rebasing while the domestic revenue used to service the debt remains unchanged. Moreover, other variations that capture debt dynamics, such as changes in interest rates, inflation, exchange rate exposures, economic growth, and fiscal and current account deficits impact the country's debt-carrying capacity over and above the debt-to-GDP criterion. Likewise, there are other underlying financing risks, such as maturity mismatch (i.e. borrowing short term to finance large projects with long gestation periods for which little or no revenue is available to service the debt) and currency mismatch (borrowing in foreign currency while revenue streams are in local currency yet having to repay the debt in foreign currency) (Ndulu 2018). Beyond actual debt levels, debt management, the prudence of fiscal policy, the quality of public investment, and the integrity of fiscal institutions also matter.

Based on the above background and in view of rapidly rising debt accumulation, there is a need to undertake in-depth analysis of public debt in Tanzania and the implications for future sustainability of debt. The objective of this paper is to analyse public debt and debt sustainability, focusing on external debt. Besides exploratory analysis of the evolution and structure of debt, debt sustainability is examined on the basis of the recent debt sustainability analysis (DSA) undertaken by the government of Tanzania using the IMF-World Bank debt sustainability framework (DSF), complemented by empirical estimation of a fiscal reaction function, which is an approach to

¹ This is, however, still lower than the 90 per cent median debt-to-GDP ratio for African countries before the Heavily Indebted Poor Countries (HIPC)/Multilateral Debt Relief Initiative (MDRI) period.

² These include Chad, Mozambique, South Sudan, and Republic of Congo.

analysing debt sustainability commonly used in the empirical literature (Bartoletto et al. 2013; Paret 2017; Tóth 2011).

The rest of the paper is as follows: Section 2 provides a synopsis of the historical background of debt and debt relief initiatives in Tanzania, followed by an overview of recent macroeconomic indicators and the evolution of debt in Section 3. A review of empirical literature on debt sustainability is given in Section 4. Section 5 provides a discussion on the conceptual framework of debt sustainability in the context of low-income countries, while Section 6 provides an overview of debt sustainability frameworks. A discussion of the results of the DSA based on the IMF-World Bank framework and the estimation of a fiscal reaction function are discussed in Section 7. Section 8 provides conclusions and policy implications.

2 Historical background: debt and debt relief initiatives in Tanzania

Tanzania has been one of the major recipients of foreign aid and concessional loans, mostly to fund the state-owned enterprises that dominated practically all sectors of the economy after independence.³ According to Nord et al. (2009), by 1988 there were about 400 parastatals, accounting for 20 per cent of GDP. Most of the enterprises could not yield the intended results due to mismanagement and inefficiency, and instead drained the meagre domestic resources, while some plunged into heavy debt.⁴

Thus, by the mid-1980s the country was already struggling with mounting debt service (Figure 1), compounded by low export earnings and the impact of external shocks, including the oil price shocks of the 1970s and poor weather. Real GDP growth averaged 2.3 per cent during the period 1981–85. The public sector was bloated, while the domestic revenue base was limited.



Figure 1: External sustainability indicators, 1970–2007

Source: reproduced from Nord et al. (2009: 2, Figure 2), with permission.

³ These were mainly created in a bid to promote self-reliance and collective ownership of resources in the spirit of socialism as articulated in the Arusha Declaration of 1967 (Nord et al. 2009).

⁴ These and other challenges that had crippled the economy by the early 1990s are documented in former president Benjamin William Mkapa's memoir *My Life, My Purpose: A Tanzanian President Remembers* (2019).

As the economic crisis deepened, various economic reforms were initiated in the mid-1980s towards transforming the economy from state control to an open market economy. However, substantive reforms were witnessed in the early 1990s, including the liberalization of the economy and the foreign exchange market. Fiscal reforms including fiscal management to constrain government spending and tax policy reforms, such as the introduction of VAT in 1998 to mobilize domestic revenue, were also undertaken.

The economy was heavily dependent on foreign aid from development partners, especially the Nordic countries and international financial institutions. By the early 1990s, foreign aid was funding about 60 per cent of the development budget (Calderisi 2015). The public sector had also grown faster, leading to budgetary challenges. The average public external debt as percentage of GDP and public external debt service as a percentage of exports exceeded 100 per cent and 30 per cent, respectively, in the period 1986–95 (Figure 1). The debt-to-GDP ratios reached 122 per cent in 1988 and 179 per cent in 1994 (Mollel 2009). A weak revenue base, a shortage of foreign exchange especially in the period prior to economic liberalization, and the transfer of parastatal debt obligations to an already over-stretched government budget led to a build-up of external payment arrears.

The government needed reprieve from the heavy debt burden,⁵ and hence it was not surprising that Tanzania was among the first countries to get debt relief under the HIPC Initiative. This initiative was launched in 1996 by the IMF and the World Bank together with other multilateral creditors as the first comprehensive effort to address unsustainable debt in the world's most heavily indebted poor countries. In October 1999, the initiative was enhanced by increasing the number of eligible countries and raising the amount of debt relief. Tanzania reached 'decision point' in April 2000 and 'completion point' in November 2001, after implementing a raft of macroeconomic and structural reforms, including coming up with a 'Poverty Reduction Strategy Paper' (PRSP), thus becoming the fourth country to reach completion point after Bolivia, Mozambique, and Uganda.⁶ The structural reforms included a reduction of the role of the state in the economy through privatization and liberalization of the economy. By December 2000, 334 privatizations had been completed.

At the completion point, the IMF estimated that debt service relief from Tanzania's creditors under the enhanced HIPC Initiative amounted to approximately US\$3 billion over time (\$2.026 billion in net present value/NPV terms) as of 2001. As a result, the NPV of Tanzania's total external debt was reduced by about 54 per cent (IMF 2001). The NPV of the debt-to-export ratio was expected to remain well below the target ceiling of 150 per cent throughout the period 2000–20, with a further fall in debt ratios after taking into account additional bilateral assistance beyond the enhanced HIPC Initiative (IMF 2001).

Besides the HIPC Initiative, Tanzania also benefited from the MDRI. To complement the HIPC Initiative, the MDRI was launched in June 2005 following the G8 major industrial countries' proposal for the multilateral institutions—i.e. the IMF, the International Development Association (IDA) of the World Bank, and the African Development Fund (AfDF)—to cancel 100 per cent of their eligible debt claims on countries that had reached, or would eventually reach, the HIPC

⁵ At the end of 1998, external debt which stood at US \$ 5 billion was larger than Tanzania's estimated GDP.

⁶ To qualify, eligible countries needed to demonstrate the capacity to prudently use the assistance granted by establishing a satisfactory track record under programmes supported by the IMF and the International Development Association (IDA) (decision point) and implementing a poverty reduction strategy together with a set of growth-enhancing measures (interim relief), followed by completion point—the stage at which a country became eligible for full and irrevocable debt relief.

completion point. The total committed debt relief for Tanzania from both the HIPC and the MDRI frameworks as at end of August 2018, amounting to over US\$6 billion, was comparatively higher than that of HIPC peers such as Uganda, Ethiopia, Malawi, and Rwanda (Figure 2).



Figure 2: Committed debt relief (HIPC and MDRI) as of end August 2018 (US\$ millions)

The debt relief initiatives substantially alleviated Tanzania's debt burden, leading to a significant decline in debt indicators. Debt stock was reduced to below US\$5,000 million in 2006, while the ratio of debt to gross national income (GNI) and debt as a percentage of exports declined to as low as 33.6 per cent and 128.7 per cent, respectively (Kaiser et al. 2009). Debt service as a percentage of exports declined to 3.4 per cent by 2006. The ratio of public external debt to GDP declined by more than half from 37 per cent in 2005/06 to 16.6 per cent in 2007/08 (Nkhata 2009). Similarly, the public external debt stock was cut nearly by half to \$4.69 billion in 2006/07. Debt service payments were cut substantially from about \$193 million in 1999/00 and \$121 million in 2000/01 at the completion point to as low as below \$50 million a few years thereafter, before starting to rise markedly especially from 2012. Debt service as a percentage of government revenue declined from 19 per cent in 2000/01 before the HIPC Initiative to an average of 7.7 per cent during the period 2000/01–10/11.

The debt relief initiatives not only lessened the debt burdens of recipient countries but also acted as a catalyst to economic growth. This is supported by a study by Cassimon et al. (2015) which showed that debt relief, especially the enhanced HIPC Initiative, had had a positive impact on recipient countries' total domestic revenue and public investment. Most of the HIPCs used the breathing space to accumulate more debts, some of them from non-concessional sources. Consequently, recent data indicate that debt service burdens in post-HIPCs, including Tanzania, are on average rising.

3 Recent macroeconomic indicators and the evolution of post-HIPC debt

3.1 Recent macroeconomic indicators

Tanzania is one of the few countries in the region that have registered a notably high real GDP growth rate, averaging 6–7 per cent a year over the last decade—no mean achievement for a low-income country. That notwithstanding, concerns have been raised regarding the quality of growth

Source: authors' compilation based on IMF (2019).

in terms of economic transformation and inclusivity. While the poverty rate has declined, the absolute number of poor citizens has not, given the high population growth rate (World Bank 2019). The decline in the current account deficit from as high as 10 per cent in 2014 to 3.5 per cent of GDP by 2018 was partly aided by a decline in oil imports (Table 1). In terms of composition, export growth has stagnated. The export of goods and services as a percentage of GDP is about 17.2 per cent on average.

Tanzania's fiscal deficit has remained relatively low, which could be an indication of effective management of public spending. However, the fiscal deficit increased to 4.3 per cent of GDP in 2017/18 from 1.5 per cent in the previous fiscal year. The relatively lower deficit in 2016/17 reflected a slow budget execution, and delays in external financing and project preparation and implementation (IMF 2018b).

Indicator	2014	2015	2016	2017	2018	2019**
Real GDP growth	6.7	6.2	6.9	6.8	7.0	6.3
Public debt to GDP (%)	32.6	33.4	37.1	39.6	38.5	38.2
National debt to GDP* (%)	37.7	38.8	43.8	46.4	46.1	48.6
External debt to GDP (%)	39.0	30.6	35.7	34.6	35.0	37.5
Total investment to GDP(%)	37.6	32.7	32.1	34.0	38.9	37.5
Exports to GDP (%)	17.5	19.9	17.7	16.8	17.0	17.3
Imports to GDP (%)	27.6	25.3	19.7	19.1	20.2	20.4
Current account balance to GDP (%)	-10.0	-7.9	-4.2	-2.9	-3.5	-3.2
Fiscal deficit to GDP (%)—FY	-2.7	-3.5	-1.5	-4.3	-1.3	-2.5
Domestic savings to GDP (%)	24.3	25.3	30.0	30.2	30.6	33.4
GDP per capita (US\$)	1,030	947	966	1,002	1,039	1,104
GDP per capita (constant) growth	3.6	3.0	3.7	3.6	3.8	3.2
External debt stocks % of exports	165.8	179.4	204.4	221.2	238.7	256
Population (millions)	48.5	49.9	51.5	53.1	54.6	56.3

Table 1: Recent macroeconomic indicators

Note: * including private debt; ** estimate; FY = fiscal year.

Source: authors' compilation based on IMF and Ministry of Finance and Planning data and World Bank (2020b).

Tax revenue as a percentage of GDP has remained more or else constant, averaging about 11 per cent over the period 2000–17, which is still relatively low by regional standards (Figure 3). Although there has been a slight increase in recent years following the various tax reforms undertaken to enhance tax compliance and administration, the tax base is still narrow. Domestic revenue mobilization is critical in terms of the ability or capacity to service and pay debt, but more importantly, as a revenue base for funding the much needed growth and development of the economy.





Source: authors' compilation based on ICTD and UNU-WIDER (2019).

Though the country has made notable economic progress, Tanzania's income per capita is still low, with an estimated nominal GDP per capita of US\$1,104 in 2019 (Table 1). Tanzania's development vision is to become semi-industrialized by 2025. However, the country has had to borrow in a bid to meet the development goals. This has mainly been undertaken through external borrowing, given the relatively thin domestic financial market.

Thanks to the debt relief initiatives, there has been an ample space for public borrowing over the years, leading to a rapid accumulation of debt, especially in recent years. Public debt as a percentage of GDP rose to 38 per cent as at end 2016/17 from 21 per cent about a decade before that. Total national debt including private debt increased to 48.6 per cent in 2019 from 37.7 per cent in 2014. External debt stock as a percentage of exports increased by 54.4 per cent over the same period, to 256 per cent. However, given the rising costs of servicing debt coupled with the relatively low domestic tax revenue base and saving rate, and in light of dwindling foreign aid inflows,⁷ the country faces a challenge in revamping development in terms of the huge investments in infrastructure and human capital that are needed to meaningfully grow and create jobs for the increasingly youthful population, while maintaining debt sustainability.

3.2 Trends and structure of Tanzania's post-HIPC debt

National debt stock, comprising external (public and private) and domestic debt, has evolved over time, largely driven by public sector debt. Total national debt increased to US\$28.4 billion at the end of June 2019 (about 49 per cent of GDP), from US\$7.6 billion at the end of June 2008 (31 per cent of GDP), an average increase of 24 per cent annually over the decade. In particular, external debt (public and private) also recorded an increasing trend during the last decade reaching US\$21.9 billion (37.8 per cent of GDP) with an annual average growth rate of 12 per cent (Figure 4). As at end of December 2019, national debt stood at US\$28.7 billion (Bank of Tanzania 2020).

⁷ External grants dropped from 5.7 per cent of GDP in 2004/05 to 1.2 per cent of GDP in 2014/15 (IMF 2018b).

Public sector debt has remained dominant, accounting for an average of 78 per cent of total external debt. Worth noting, the private sector external debt doubled to US\$4.2 billion as at end of June 2019 from US\$2.1 billion at the end of June 2010, an annual average increase of 8 per cent.



Figure 4: Composition of external debt by borrower (US\$ millions)

Evolution of public debt

Both external and domestic debt have been increasing in the post-HIPC period, with external debt accounting for a significantly higher proportion of public debt—73.7 per cent as at December 2019 (Figure 5). Although domestic debt still accounts for a relatively smaller proportion, it had increased steadily to 14,435.2 billion Tanzanian shillings (TZS) at the end of December 2019, with treasury bonds and stocks accounting for 79.6 per cent (Bank of Tanzania 2020).



Figure 5: Trend and composition of public debt (US\$ millions)

Source: authors' compilation based on debt series from Bank of Tanzania Monthly Economic Reviews.

Tanzania's total public (domestic and external) debt stood at US\$22.5 billion as of June 2018 (38.7 per cent of GDP in nominal terms), up from \$6.1 billion (24.5 per cent of GDP) as at end of June

Source: authors' compilation based on Bank of Tanzania Annual Reports.

2008, representing an average increase of 4 per cent per annum during the period. External debt increased to \$15.8 billion as at end of June 2019 from \$4.4 billion at the end of June 2008, which represents an annual increase of 13 per cent over the decade. Growth in debt is mainly attributed to increased borrowing to finance public infrastructure projects, particularly those related to transportation, natural gas, and power generation and transmission.

External debt by concessionality

The rising financing needs to fund development projects have come at a time when the financing landscape is significantly changing. Traditional and relatively concessional sources of financing have been declining, giving rise to new non-concessional financing sources that are more complex, costly, and risky. Consequently, the proportion of non-concessional debt has been rising over time, reaching 41.8 per cent of total external debt by the end of June 2019, from below 10 per cent as at end of June 2011(Figure 6).





Source: authors' compilation based on debt series from Ministry of Finance and Planning and Bank of Tanzania.

Sources of external debt by type of creditor

Tanzania's external debt was traditionally linked to concessional sources—that is, multilateral and bilateral creditors. However, the post-HIPC period has been characterized by a gradual decline in the proportion of debt held by multilateral and bilateral creditors from 54.7 per cent and 17.7 per cent as at end of June 2012 to about 46.6 and 9.4 per cent as at end of June 2019, respectively. On the other hand, the period witnessed an increasing proportion of debt from commercial sources, from 18.0 per cent to 33.4 per cent (Figure 7). This has implications on debt servicing costs, given that the latter is relatively costlier.



Figure 7: Composition of external debt by creditor category (%)

Source: authors' compilation based on debt series from Bank of Tanzania Annual Reports.

Currency composition

The currency composition of public external debt shows that the debt portfolio is most exposed to the US dollar, followed by the euro, which is largely a reflection of the historical financing pattern, the international financial set-up, and the external trade pattern of the country (Table 2). Traditionally, the country has accessed more financing from international financial institutions using basket currencies, i.e. Special Drawing Rights and African Unit of Accounts, of which the US dollar is the dominant currency. The emergence of new lenders has not changed the currency composition—for instance, most of the financing from the exim banks of China, India, and South Korea have continued to be denominated in USD. Hence, there is an inherent currency risk, if, for instance, the domestic currency were to depreciate significantly against the dollar.

Currency	2010/11	2010/12	2010/13	2010/14	2010/15	2010/16	2010/17	2010/18	2010/19
US dollars	40.6	44.8	49.7	52.3	56.0	55.4	55.4	55.9	55.6
Euro	34.9	32.2	28.7	27.3	24.9	25.0	21.4	21.1	22.5
Japanese yen	10.2	9.6	8.2	8.0	7.5	8.6	7.9	7.8	7.8
Great British pound	9.0	8.4	7.5	7.2	6.7	6.6	4.7	4.8	4.7
Chinese yuan	3.7	3.5	4.7	4.0	4.0	3.6	9.7	9.7	8.8
Others	1.6	1.5	1.2	1.2	0.9	0.8	0.9	0.7	0.6
Total	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0

Table 2: Currency composition of external debt (%)

Source: authors' compilation based on data from Ministry of Finance and Planning.

Maturity profile of external debt

The remaining maturity structure of public external debt as at end of June 2019 shows that more than half of the debt (56.8 per cent) will mature only after ten or more years, which reflects the significant portion of concessional debt in the portfolio (Figure 8). Based on the maturity profile, it can be argued that central government external debt is relatively less exposed to refinancing risk. However, the financing risks and challenges associated with the notable share of external debt of less than ten-year maturity cannot be ignored.





Source: authors' compilation based on data from Ministry of Finance and Planning.

External debt by use of funds

The profile of external debt by use of funds as at end of June 2019 indicates that most of the debt contracted was utilized for budget and balance of payments support, energy and mining, agriculture and industrial development, and transport and construction activities (Figure 9). The financed projects have already started to yield results in terms of significant improvement in transportation activities and economic growth.



Figure 9: Composition of external debt by use of funds as at end June 2019

Source: authors' compilation based on debt series from Bank of Tanzania's Annual Reports.

3.3 Regional comparison

Despite the notable increase in debt in the post-HIPC and MDRI frameworks, the country's debt has been relatively low regionally. For instance, the solvency debt indicators in terms of the ratio of nominal and present value of external debt to GDP show that Tanzania's debt is below that of Mozambique, Zambia, and Kenya, though slightly higher than Uganda's (Figure 10). The present value (PV) of public debt to GDP is, however, comparatively higher than one for Kenya and Ethiopia.



Figure 10: Present value of ratio of external debt and public debt to GDP (%)

Source: authors' compilation based on IMF data and World Bank (2020a).

Likewise, the liquidity indicators show that Tanzania's external debt-to-exports ratio is below that of some of her peers and neighbouring countries. However, in terms of public debt to exports, the ratio is slightly higher than that for Ethiopia, Kenya and Uganda (Figure 11).



Kenya

Figure 11: Present value of ratio of external and public debt to exports (%)

Source: authors' compilation based on IMF data and World Bank (2020a).

Mozambique

Zambia

Ethiopia

Tanzania

Uganda

Figure 12 shows that whereas the ratio of external debt to GNI for Tanzania declined drastically relative to that of its peers after the cancellation of debt in early 2000s, the cost of external debt servicing has recorded a rising trend, with a sharp increase since the early 2010s. Though it is lower than Kenya's, the latter is not an HIPC and is classified as a lower-middle-income country.

Tanzania

Ethiopia

Zambia

Mozambique

Kenva

Figure 12: Comparative external debt to GNI and debt service payments



Source: authors' compilation based on World Bank (2020a).

4 Literature review

In the literature, the standard approach to debt sustainability analysis focuses on the debt-to-GDP ratio. Public debt has the potential to increase liquidity in the economy, enhance consumption patterns, and improve citizens' wellbeing, which ultimately decreases the burden of borrowing on households. At the same time, the escalation of debt to higher levels can compromise a country's welfare, especially if it crowds out private investment and leads to lower output (Chatterjee et al. 2017). Analysis by Melina et al. (2016) shows that even for resource-rich developing countries, expanding public investment without putting in place saving or borrowing strategies can expose the economy to debt sustainability risks, particularly if public investment is inefficient or the anticipated future resource revenues are not realized.

Bassanetti et al. (2016) show that debt distress can be predicted on the basis of debt dynamics that go beyond actual debt levels-a declining debt-to-GDP ratio is often associated with less likelihood of debt distress, particularly under the scenario in which the debt ratio is rising. Using a probabilistic approach to predict the public debt of six emerging countries from different economic regions (Russia, Turkey, Argentina, Brazil, South Africa, and the Philippines), Paret (2017) found that fiscal policies do not necessarily act pre-emptively to prevent the debt ratio from rising. To assess the macro-fiscal linkages, a fiscal reaction function was estimated with a view to analysing how primary balance responds to debt ratio and output gap. The coefficient of the lagged debt ratio was positive and significant, indicating that countries' fiscal policy was responsive to debt. However, the response to the output gap was positive but statistically insignificant. Further analysis suggested that those emerging countries with strategies that promote fiscal responsibility have lower primary deficits or small surpluses. Countries that have accrued primary surpluses (the Philippines and Turkey) demonstrated the application of more sustainable debt strategies compared with those burdened by primary deficits (Argentina, South Africa, Russia, and Brazil). Consequently, strong fiscal responsiveness to debt (fiscal tightening whenever debt increases) is bound to enhance debt sustainability.

Makin and Pearce (2014) examined the sustainability of public debt at the subnational level in Australia and found that almost all administrative units such as states and general government territories had witnessed unstable debt levels. The study further showed that the capacity to stabilize debt depended on the cost of debt servicing and the size of public debt relative to gross state product (GSP), as well as the size of the primary budget balance. To improve the debt situation, all subnational governments were expected to transform primary deficits into surpluses.

Beqiraj et al. (2018) investigated government responses to debt accumulation using panel data for 21 dissimilar Organisation for Economic Co-operation and Development (OECD) countries. Variables utilized included cyclically adjusted primary balances, debt-to-GDP ratio, debt-to-potential-GDP ratio, output gap, and an indicator for business cycle. The study showed a close relationship between debt and structural primary balance, thus reaffirming the conclusion that unrestricted growth in the debt-to-GDP ratio for long periods can be detrimental. Structural primary balance also responded positively to transitory changes in GDP.

Mahmood et al. (2009) used traditional threshold debt ratios and theoretical models to assess the debt sustainability level of Pakistan over the 1975–2007 period. The study found that the debt sustainability challenge was precipitated by current account and fiscal imbalances. Similarly, Jayme (2001) investigated the sustainability of the external debt in Brazil based on Johansen cointegration tests. The results showed that in the long run, deficits in the current account and external debt sustainability between 1970 and 2009 by examining primary surplus in relation to the debt-to-GDP ratio. However, the study could not establish concrete or precise evidence in support of public debt sustainability.

Using stochastic simulation methods, Karam and Hostland (2006) established probability projections of the public debt burden in the medium term with a view to assessing debt sustainability in emerging market economies. Public debt vulnerability to severe economic shocks was measured by assessing indicators such as volatilities in outputs, financial fragilities, and internal reactions of the risk premium and abrupt ceasing of private capital flows. External debt vulnerability was found to be sensitive to the exchange rate determinants as well as to prices of tradable goods. The results showed that in the medium term, the debt burden could be prevented from rising by adopting appropriate fiscal policies. Moreover, it was noted that factors such as volatilities in output and financial fragilities—especially over-dependence on short-term currency borrowing and sudden stops in private capital flows—matter.

Bökemeier and Stoian (2018) investigated debt sustainability in ten countries in Eastern and Central Europe by estimating a fiscal reaction function. The results showed that as of 2015, the debt ratios for Bulgaria and Romania were not sustainable. Moreover, linear trend analysis over a 20-year period showed volatile debt dynamics for Romania and Latvia. Comparisons between the stable debt ratios and the historical averages indicated sustainability of debt for most countries, except Bulgaria in the long run. However unclear results were reported for five countries—Slovenia, Czech Republic, Poland, Hungary, and Slovakia—indicating limitations of the model employed.

Llorca (2017) examined a cohort of 24 emerging and developing economies in Asia to assess external debt sustainability from 1993 to 2014. The study used the PV methodology to assess whether the external debt of countries—which were further grouped into the four regions of South-East, South-West, Central, and the Pacific—was sustainable in the long run. External debt was found to be sustainable for all the economies assessed.

Pradhan (2019) examined changes in the status of the external debt and externally induced vulnerability indicators in India. Notably, the size of the external debt of India had increased but was constant, at about 20 per cent, as a ratio of GDP. The rise in the external debt was attributed to borrowing by non-government sectors. India's vulnerability to externalities with regard to reserve adequacy indicators was noted to be comparatively low.

Ncube and Brixiová (2015) examined public debt sustainability in Africa over the 2008–12 period based on a debt-stabilizing primary-balance framework. The two main approaches to debt sustainability utilized were: (i) the IMF and World Bank approach, which focuses on the projection of the debt path in relation to given thresholds, and (ii) the debt-stabilizing primary-balance approach, which identifies primary balances needed to achieve a given debt path, based on various assumptions about changes in real interest rate and growth. A positive primary-balance gap⁸ would imply a rise of the debt-to-GDP ratio over a period of time without fundamental fiscal policy adjustment. The study found that in half of the countries examined, primary balances were above those needed to keep the level of public debt at 2007 debt levels, and in most cases above those required to ensure public debt sustainability. Sustainability was driven by the interest rate–growth differential (IRGD), accentuating the significance of growth or borrowing to enhance growth. Since in the long run IRGDs are likely to narrow, the study underscored the greater role of fiscal policies. In other words, African policy-makers needed to adopt sound fiscal policies while pursuing growth-enhancing investment, including through borrowing.

Using historical time series data spanning the period 1865–2010, Naraidoo and Raputsoane (2015) examined debt sustainability in South Africa, taking into consideration the possibility of nonlinearities in the form of threshold behaviour by fiscal authorities based on recent history of indebtedness as well financial crises. The results showed a debt-to-GDP threshold ratio of about 56 per cent. The results further revealed that previous debt levels played a fundamental role in informing debt adjustment decisions. There was evidence in support of the need for fiscal sustainability, such that a rise in debt as a ratio of GDP beyond the threshold results in stronger fiscal consolidation.

Using data for the period 1970–2011, Kayandabila and Manyama (2013) concluded that the fiscal policy for Tanzania was unsustainable over the study period, based on analysis undertaken using unit root tests and cointegration tests. In his assessment of the main risks to public debt sustainability in post-HIPC countries Malawi, Uganda, and Tanzania, Nkhata (2009) noted the vulnerability of external debts to increased lending rates in these countries, and concluded that the acquisition of additional debt should be carefully analysed to ensure it is sufficiently concessional to promote future debt sustainability. The stress tests under the DSA analysis showed Tanzania's total public debt to be sensitive to low GDP and worsening primary balance.

Other than the DSA that is commonly undertaken by the IMF and the respective authorities in low-income countries, empirical analysis of debt sustainability in post-HIPC countries like Tanzania is limited. Most of the empirical literature on debt sustainability has focused on emerging and developing market economies. Most of the studies employ time series analysis that entails the estimation of a fiscal reaction function, with a general emphasis on the role of fiscal policy in ensuring debt sustainability. In terms of empirical analysis, we follow a similar approach by estimating a fiscal reaction for Tanzania.

⁸ The primary-balance gap is the difference between the stabilizing and the actual primary balance.

5 Conceptual framework

Assessing debt sustainability is complex, particularly given the dynamic nature of debt and the varied factors that drive both its accumulation and a country's ability to pay in the short run and long run. There is no universally accepted definition of sustainable debt, though there is a general consensus that exploding debt is not sustainable. Low or high debt levels do not necessarily imply that debt is sustainable or unsustainable in the long run. There are several factors that need to be considered when conceptualizing and considering public debt and its sustainability, especially in the context of developing African countries like Tanzania. These are briefly discussed below.

5.1 Debt sustainability as a necessary condition but not an end in itself

From a developmental perspective, borrowing, whether government or private, is an important avenue for financing the investment that is critical to achieving sustainable development goals. The role of government in the provision of public capital through public investment such as infrastructure is well acknowledged in development literature.

On one hand lies the historical Keynesian argument that scaling up public investment via government borrowing, while it may lead to an increase in the debt ratios in the short run, crowds in private investment and boosts export growth, hence spurring higher growth in the medium to long term and eventually lowering the debt ratios over time. Therefore, what matters is the sustainability of long-run growth and development, for which debt sustainability is a necessary condition and not an end in itself (Pinto 2018). However, this is based on the implicit assumption of efficient and productive public spending: efficiency and absorptive capacity play a role in determining the ultimate impact of public investment on growth, and hence an economy's ability to pay (Mustapha and Prizzon 2015).

At the macro level, the need for government borrowing follows from the standard macroeconomic identity:

$$Y \equiv C + I + G - T + X - M \tag{1}$$

where: Y is national output, C consumption, I investment, G government expenditure, T taxes, X exports, and M imports.

Equation 1 can be rearranged to obtain the following identity:

$$S - I \equiv G - T + X - M \tag{2}$$

where S is savings, S - I is net savings, G - T represents budget deficit after into-account transfers, and X - M is the trade surplus (net exports).

From the identity in Equation 2, the excess of saving over investments (S - I) of the private sector equals the government budget deficit plus the trade surplus (the economy's external balance). Any sector in the identity that spends more than it receives must borrow to finance excess expenditure. Borrowing can arise when S - I < 0; X - M < 0; and G - T > 0.

The value of access to foreign credit to finance long-term growth depends on the development status of the country and availability of domestic funds (Stiglitz 2016). For most African countries the saving rates are low and capital is often in short supply, compared with developed or East Asian countries where the saving rate is relatively higher. That notwithstanding, short-term lending

could be of limited benefit compared with foreign direct investment (FDI) that brings with it access to new technologies and markets, entrepreneurship, and capital (Stiglitz 2016).

On the other hand, high debt burdens can impede growth and sustainable development, and could lead to debt overhang.⁹ If the cost of debt servicing becomes overly high, it could implicitly lead to higher taxes due to the government's need to collect more revenue in order to meet external debt obligations, and could thus act as a disincentive to domestic investment (Mustapha and Prizzon 2015). There is a vast literature on the negative effect of debt overhang on investment that particularly emerged after the debt crisis of the 1980s in developing and low-income countries (see Sundell and Lemdal 2011).

5.2 Debt dynamics matter

The critical level of debt depends on the 'state of nature', i.e., a variety of factors that determine the prospects of a country (Stiglitz 2016). Although the country may be viewed today as being far below debt limits, its exchange rate may suddenly deteriorate, interest rates could rise, and/or GDP may fall, leading to an increase in the (domestic) value of debt and lowering the country's debt capacity. In this context, it is not only the debt levels that matter but also the debt dynamics. Debt levels provide an incomplete picture. The debt dynamics are closely linked to structural vulnerability and exposure to global shocks, such as volatility in commodity prices, which can adversely affect export earnings and the ability to service the debt, especially for a less diversified economy.

Solvency and liquidity debt indicators are commonly used in debt sustainability analysis. Solvency risk refers to the capacity and overall viability of a country's economy to accommodate and meet debt obligations—measured by debt-to-GDP ratio relative to a given threshold. Liquidity risk reflects a country's ability to manage and meet debt service obligations falling due—measured as the ratio of debt service to exports and to revenue. These two sets of challenges are not necessarily congruent—a country could prospectively be in trouble in one category of risk while not in the other. It is important for the debt ratios to be considered in conjunction with the overall dynamics of key domestic and external factors that determine the actual outcome in the medium to long run.

5.3 Composition of debt

Unlike the external debt accumulation of the 1970s and 1980s (before the HIPC Initiative) among the low-income countries, which was largely due to an increase in debt to multilateral institutions, the current build-up is largely due to a significant increase in commercial borrowing. Since 2006, for instance, more than ten SSA countries including Tanzania have issued foreign currency bonds in international debt markets. Other countries that have issued sovereign bonds include Kenya, Nigeria, Senegal, Zambia, Rwanda, Ghana, and South Africa. The maturities of the bonds generally range from five to ten years. This relatively short maturity structure does not match the life of infrastructure projects with much longer gestation periods of investment returns (Coulibaly et al. 2019, Ndulu 2018).

Increased access to international capital markets has introduced new financing opportunities but also new risks, such as exposure to volatility in interest rates and international capital flows. It has also translated into higher debt servicing costs. Ndulu and O'Connell (2019) observe that development assets, which include hard-acquired improvements in institutions, infrastructure

⁹ This is a situation in which the debt burden is so large that earnings generated by new investment are entirely appropriated by existing debt holders, in which case the debt burden leads to an implicit tax on domestic capital.

projects nearing fruition, and human capital formation are at stake. Debt servicing costs, as well as the design features of debt contracting and the creditor structure, matter to a country's ability to achieve debt sustainability (Coulibaly et al. 2019). The risk of debt-induced fiscal distress is heightened by maturity mismatches, rollover risks, currency mismatch, and coordination challenges in debt restructuring, especially given a plurality of creditors, reflecting the increase in private creditors and non-Paris Club members (Ndulu and O'Connell 2019).

In general, the region's bilateral debt has increasingly shifted towards non-traditional lenders such as China, India, and South Korea, away from the Paris Club member countries. In 2017, the Paris Club countries owned less than 7 per cent of Africa's outstanding external debt, compared with 15 per cent in 2008 (Coulibaly et al. 2019). With most of the debt denominated in foreign currency, the depreciation of domestic currencies has a negative effect, since most of the government revenues are in local currency, unless exports are substantively boosted to counter or pay off the debt burden.

Also worth noting is the distinction in the risk profile between domestic and external debt. Whereas high demand for domestic debt can have repercussions on local interest rates if financial markets are relatively thin or not well developed, it is easier to deal with the latter in terms of rollover and less exposure to exchange rate volatility.

5.4 The denominator matters

Whereas the focus is often on the numerator (debt), and perhaps the quantity of the denominator (GDP), the composition and quality of the denominator matters. For instance, from the aggregate demand identity in Equation 1, export (X) growth is important as a key source of foreign exchange earnings, given that foreign debt is mostly contracted and paid in foreign currency. On the domestic front, the ability to generate domestic revenue is important, the key component being domestic tax (T) revenue in most cases.

5.5 Macroeconomic and public debt management

The sustainability of public debt can be interpreted as the outcome of the interaction of fiscal policy with the economic environment, which goes beyond a statistical concept as often depicted in the recent literature. If debt is not to explode over time, policy-makers have to respond to the changing conditions in the macroeconomic environment (Collignon 2012). Debt levels can be low but, if not prudently managed, could still lead to accumulation and sustainability challenges. Public debt management entails establishing and executing a strategy for managing the public debt in a way that raises the required amount of funding at the lowest possible cost over the medium to long term, while taking into consideration exposure to market factors such as exchange and interest rate volatilities (IMF 2014).

Moreover, as countries continue to witness a rise in contingent liabilities, particularly after the global financial crisis, a consolidated fiscal account that takes into account explicit and implicit government contingent liabilities is critical in the assessment of exposure to debt risk. The situation could get worse in the aftermath of the COVID-19 pandemic.¹⁰

¹⁰ The novel virus that originated in Wuhan in China was declared a global pandemic in March 2020.

6 Methodology: public debt sustainability assessment frameworks

Besides exploratory analysis of the evolution and structure of debt, debt sustainability is examined based on the commonly applied debt sustainability framework developed by IMF and World Bank and an empirical estimation of a fiscal reaction function for Tanzania.

6.1 IMF-World Bank debt sustainability framework

The standard approach to DSA focuses on the debt-to-GDP ratio. The IMF-World Bank Debt Sustainability Framework for Low-Income Countries (LIC DSF) and Debt Sustainability Framework for Market Access Countries (MAC DSF) are the most commonly used frameworks for assessing and conducting DSA. The frameworks were introduced in 2005 and have since undergone various developments, with the latest review in July 2018. They are widely used by the Bretton Woods institutions, lenders, and borrowers as the main basis for providing guidance for lenders' borrowing and creditors' lending decisions.

The DSA consists of an analysis of a country's projected debt burden over the next ten years and its vulnerability to shocks, based on medium-term projections of macroeconomic variables and assumptions regarding changes in the primary balance, on the basis of which baseline and stress tests are calculated. The assessment of the risk of external and overall debt distress is based on the debt burden thresholds and benchmarks (Table 3). The framework focuses on the present value of debt obligations for comparability. A 5 per cent discount rate is used to calculate the PV of external debt. To assess debt sustainability, debt burden indicators are compared with indicative thresholds over the projection period. The risk of external public debt distress is based on four ratings: low risk, moderate risk, high risk, and in debt distress. The key advantage of the DSF/DSA framework is its simplicity, which allows it to be easily replicated for different countries.

	PV of external debt in % of		External deb	t service in % of	PV of total public debt in % of		
	GDP	Exports	Export	Revenue	GDP		
Weak	30	140	10	14	35		
Medium	40	180	15	18	55		
Strong	50	240	21	23	70		

Table 3: Debt burden thresholds and benchmarks under the DSF

Source: reproduced from IMF (2020).

The DSF, though heavily relied on, has received its own share of criticisms based on its limitations. For instance, one of the criticisms is that the focus on the present value of debt is outdated. Moreover, only concessional external debt is discounted, at an arbitrary discount rate of 5 per cent. According to Pinto (2018), given the profusion of different types of public debt—concessional, market debt, Eurobonds, and non-concessional bilateral loans—at different interest rates, today, nominal debt makes more sense, and it is the weighted average of these different types of public debt that is relevant for debt dynamics. Additionally, the IMF-World Bank framework has been found to be overly optimistic with respect to projections and assumptions about growth and fiscal adjustments, yet borrowing agreements are based on DSA obtained using the framework, which means that countries keep borrowing (Atingi 2019). Analysis by Atingi (2019) showed a tendency to optimism bias, with significant errors in the growth forecasts for some countries. Moreover, the fact that simulations do not take into account the uncertainty of macroeconomic forecasts given the volatile global environment faced by low-income and emerging market economies has been noted as a major drawback (Paret 2017). Additionally, the DSF does not distinguish debt

sustainability that arises due to misuse of public resources and weak fiscal institutions from that linked to large but relatively more meaningful investments in infrastructure.

Concerns have also been raised regarding the need for country-specific benchmarks for domestic debt and total public debt beyond internationally set benchmarks for assessing sustainability and vulnerabilities. Moreover, robust analysis of variables that drive debt relative to GDP, fiscal deficits, interest rate differentials, exchange rate risks, and current account deficits is critical.

6.2 Fiscal reaction function

Governments face a PV borrowing constraint in the sense that they must intertemporally balance their budgets by setting the current market value of debt equal to the discounted sum of expected future surpluses. A violation of intertemporal budget balance can be taken as an indication that the fiscal policy is not sustainable in the long run, since the value of debt would explode over time at a rate faster than the growth rate of the economy (Collignon 2012). This, however, does not necessarily mean that budgets must be balanced.

The empirical literature focuses on testing the sustainability of the intertemporal budget constraint through the use of univariate and multivariate techniques, particularly focusing on tests for the presence of unit roots and fiscal rules (Bartoletto et al. 2013). Among the most influential works in this context include Trehan and Walsh (1988) and Bohn (1998). However, Bohn (2007) cast doubt on the necessity of stationarity and cointegration restrictions. The definition of sustainability based on the PV budget constraint has also been criticized as being ad hoc, with little economic argument (Bohn 2007).

Following Bohn (1998), a model-based approach of testing sustainability by estimating an appropriate policy rule, i.e. a reaction function for the primary surplus to the debt-to-GDP ratio, is recommended. This representation of debt sustainability refers to a relationship between public debt and primary surplus, such that when the debt rises, a sustainable fiscal policy requires an increase in primary surplus. The essence is to capture the relationship between a fiscal instrument (primary balance) as an indicator of changes in economic policy and debt stability which reflects the fiscal goal.

In line with Bohn (1998), if the government reacts quickly and efficiently to changes in government debt through primary balance, the danger of government debt becoming unstainable is averted. Based on fiscal reaction analysis, government debt is considered to be sustainable if past evidence shows that the budget improves in response to a rise in government debt (Bartoletto et al. 2013; Tóth 2011).

Following Bohn (1998, 2005), a fiscal reaction function for Tanzania is estimated. Fiscal reaction function typically describes a country's fiscal response, as captured by the primary balance, to changes in debt levels and business cycle fluctuations (output gap). A positive and statistically significant fiscal response coefficient is a sufficient condition for debt sustainability. In line with similar empirical studies such as Paret (2017) and Bartoletto et al. (2013), the equation allows for smoothed adjustment of primary balance by including lagged primary balance on the right-hand side. Allowing for primary balance to depend on its past values controls for possible deficit bias and sluggish budget response. Econometrically, this also controls for serial autocorrelation. The basic equation is specified as below:

 $pb_{t} = a_{0} + a_{1}d_{t-1} + a_{2}pb_{t-1} + a_{1}og_{t} + \varepsilon_{t}$

Where pb is primary balance measured as a ratio to GDP; d_{t-1} is the debt at the end of previous period measured as ratio to GDP, og_t is the output gap at time t and ε_t is the error term.

The above equation is estimated using general method of moments (GMM). This approach is best suited to dealing with potential endogeneity. However, the ordinary least squares (OLS) estimation results are also reported for robustness. The data used for the empirical analysis cover the period 1970–2019.¹¹ Data are from statistics produced by the IMF and World Bank, Ministry of Finance and Planning, and National Bureau of Statistics.

7 Results of debt sustainability analyses

This section provides results of the debt sustainability analyses based on the IMF-World Bank DSF and empirical estimation of the fiscal reaction function. The discussion of the former is based on the analysis conducted recently by the government of Tanzania¹² using DSF.

7.2 Debt sustainability analysis based on the DSF

The IMF and the World Bank usually work with low-income countries to produce regular DSAs using DSF. The latest DSA for Tanzania is, however, not publicly available. Nonetheless, the DSA conducted by the government of Tanzania in 2019 using outstanding debt as end of June 2019 showed that, despite the notable debt accumulation in the post-HIPC period, external (public and private) debt are sustainable for all indicators under both baseline and shock test scenarios (Ministry of Finance and Planning 2019). The analysis was based on strong performance indicators following the reclassification of the country in 2018 using the composite index of Country Policy and Institutional Assessment (CPIA) and macroeconomic performance.

The risk of public external debt distress was reported to be low under baseline and shock scenarios, albeit with sensitivity to exchange rate volatilities and export shocks. According to the analysis conducted, the PV of external debt to GDP will increase moderately, from 16.3 per cent at the end of June 2020 to 18.2 per cent in June 2022, and will thereafter decline to 9.1 per cent in 2039/40 (Table 4). Likewise, the PV of external debt to export is projected to increase from 103.9 per cent in June 2022/23 and around 53.5 per cent in the next 20 years.

The projections are, however, predicated on strong GDP growth in the future and slow-down in government borrowing after completion of the major infrastructure projects under the 'Five-Year Development Plan II'.

The liquidity ratio as measured by debt service to export is projected to decline from 11.9 per cent at the end of 2019/20 to about 9.5 per cent in the medium term and further to 7.9 per cent at the end of 2039/40. Fiscal indicators suggest that the PV of public debt to GDP will increase moderately from 27.1 per cent at the end of June 2020 to 33.2 per cent at the end of June 2030 before declining to 28.2 per cent in 2039/40. Despite fluctuations during the projection horizon, the ratio remains below the threshold of 70 per cent. Likewise, the PV of debt to revenue and

¹¹ The data covers a longer period including pre-and post-HIPC periods and hence is fairly balanced. The post-HIPC time series is too short.

¹² By the Ministry of Finance and Planning in conjunction with Bank of Tanzania.

grants is projected to rise from 163.7 per cent at the end of June 2020 to 202.5 per cent at the end of June 2030, and to decline thereafter to 160.9 per cent by end of June 2040 (Table 4).

Worth noting is the fact that the DSA results show that between 2021 and 2022, Tanzania's debt profile will be characterized by maturity bunching as shown by the rising external debt solvency and liquidity indicators. While conventionally the use of appropriate sinking funds provision to smooth the process of liquidity mobilisation for bullet debt service repayment is advisable, the challenge is the availability of resources for such provision, given that the country is implementing major development projects. Of greater importance is ensuring that the borrowing is directed to projects that promote sustainable economic growth.

Apart from the conventional thresholds embodied in the DSF template, Article 3 of the Government Loans, Guarantees and Grants Act, CAP 134, has been instrumental in protecting the liquidity position of the government as the impact of each and every loan on debt service payments is evaluated against three-year historical averages of domestic revenue and export proceeds. Likewise, government borrowing is guided by the 'Medium-Term Debt Management Strategy', which is updated annually. These have been instrumental in avoiding the pitfalls that contributed to the debt crisis before debt cancellations under the HIPC Initiative.

	Threshold	2019/20	2020/21	2021/22	2022/23	2023/24	2029/30	2039/40
External DSA								
PV of debt to GDP	55	13.3	16.8	17.6	18.2	18.0	13.7	9.1
PV of debt to exports	240	103.9	105.5	110.9	116.7	115.4	83.9	53.5
Debt service to exports	21	11.9	11.1	9.5	10.3	11.2	11.7	8.1
Debt service to revenue	23	11.9	11.9	10.1	10.7	11.6	11.8	7.9
Fiscal DSA								
PV of debt to GDP	70	27.1	27.5	28.1	29.2	29.6	33.2	28.2
PV of debt to revenue	N/A	163.7	175.6	178.4	185.3	187.8	202.5	160.9
Debt service to revenue	N/A	33.4	29.7	27.5	29.3	29.9	46.8	44.8

Table 4: Debt sustainability analysis results

Source: authors' compilation based on Ministry of Finance and Planning (2019).

According to the previous IMF-World Bank DSA for Tanzania conducted in December 2017, the risk of debt distress was indicated to be low, based on the assumption of a strong economic growth path and a projected decrease in the current account deficit, among other assumptions (IMF 2018b). Nonetheless, some debt indicators were found to be sensitive to various shocks. In particular, a one-time depreciation shock was the most extreme scenario among bound tests for most of the ratios, confirming the sensitivity of the Tanzanian economy to shocks stemming from exchange rate volatility, especially for the debt-service-to-revenue indicator. Additionally, costlier terms of external finance underscored a degree of vulnerability in the debt-to-exports measure (IMF 2018b).

Besides the impact of the debt relief that substantially reduced the country's debt burden, the low debt distress risk assessment underscores the sound macroeconomic and fiscal management that has also been associated with robust growth in recent years. Nonetheless, the sensitivity to the exchange rate and export shocks noted in both recent and previous DSAs portends liquidity risks. The challenges from maturity and currency mismatches are linked to liquidity risks. In view of the rising debt accumulation and debt servicing, reliance on commercial debt could be too costly and should, hence, be used cautiously. Given the need for the scaling up of development spending to address infrastructure gaps, it will be vital to leverage more on concessional borrowing, coupled with enhanced efficiency of public investment and debt management—particularly in light of the changing structure of debt. The results generally underscore Tanzania's vulnerability to external

shocks, the importance of expanding the export base and boosting domestic revenue sources, and prudent fiscal management over the medium to long term.

The DSA outcome of low risk of debt distress should not create room for complacency, nor be misconstrued to imply that there are no significant risks that could materialize, particularly given the strong assumptions that underpin the projections, the general limitations of the DSF, and uncertainties in the global economy. The tightening of financing conditions, potential exchange rate depreciation, and vulnerability to export shocks in view of the recession in key export markets and a fragile global economy in the wake of the notable socioeconomic impact of the COVID-19 pandemic pose risks to the growth of the economy and to debt prospects.

7.2 Empirical results based on fiscal reaction function

The empirical results of the fiscal reaction function estimated using GMM are reported in Table 5.¹³ Equation 1 refers to the basic equation as previously specified, with fiscal balance on the lefthand side and lagged public debt ratio, lagged fiscal balance, and output gap as explanatory variables. In line with the empirical literature, a positive and statistically significant coefficient of lagged public debt to GDP connotes evidence of debt sustainability. Though positive, the coefficient of lagged public-debt-to-GDP ratio in the basic model was, however, found to be statistically insignificant, signifying lack of compelling empirical evidence. Equation 2 shows that the coefficient is only significant at the 10 per cent level after inclusion of a dummy variable capturing the significant liberalization reforms undertaken in 1993, which were associated with improved fiscal balance.

	GMM		OLS	6
	Equation 1	Equation 2	Equation 1a	Equation 2b
Fb(-1)	0.10*** (7.92)	1.15*** (8.15)	0.73*** (7.05)	0.70*** (7.64)
Pub(-1)	0.008 (1.24)	0.016* (1.70)	0.009 (1.00)	0.01* (1.70)
Ygap	-0.13 (-1.09)	-0.22 (-1.44)	0.01 (0.06)	0.08 (0.75)
D1993		−0.08* (−1.65)		-0.04*** (-2.81)
D2001				0.03** (2.40)
Constant	-0.005 (-0.79)	-0.006 (-0.64)	-0.01* (-1.75)	-0.02** (2.00)
Adjusted R- squared	0.55	0.60	0.58	0.70

Table 5: Empirical results—public debt

Note: *** indicates 1%, ** 5%, and * 10/% levels of significance; t-statistics in brackets; FB(-1) = lagged fiscal primary balance, Pub(-1) = lagged public-debt-to-GDP ratio, Ygap = output gap, D1993 and D2001 = dummy variables for 1993 and 2001; J-statistic (GMM eq.1) = 0.99 (0.32); GMM eq. 2 is exactly identified.

Source: authors' own analysis.

¹³ The instruments used, which include lagged output gap and exchange rate, besides satisfying the identification conditions, are consistent with those applied in related studies such as Paret (2017). For robustness, the equation was also estimated using ordinary least squares.

In general, empirical evidence of debt sustainability is weak.¹⁴ With regard to the other variables, lagged fiscal balance is highly significant, which indicates a sluggish fiscal budget response consistent with the literature. However, the output gap was found to be statistically insignificant. The OLS results depict similar outcomes.

Besides the reaction function using public (external and domestic) debt-to-GDP ratio, the reaction function is also estimated using external-debt-to-GDP ratio given the significant share of the later. The empirical results are reported in Table 6. The results generally mirror the ones reported using public debt as a ratio of GDP. Lagged external debt to GDP coefficient is statistically insignificant in equation (1) and only significant at 10 per cent level of significance in equation (2).

	GMM		OLS	
	Equation 1	Equation 2	Equation 1a	Equation 2b
Fb(-1)	1.09***	1.17***	0.75***	0.77***
	(7.73)	(7.71)	(7.21)	(7.82)
Ext(-1)	0.008	0.010*	0.005	0.007
	(1.40)	(1.74)	(1.00)	(1.45)
Ygap	-0.18	−0.21	-0.01	0.02
	(-1.57)	(−1.38)	(-0.10)	(0.11)
D1993		-0.07* (-1.65)		
Constant	-0.003	-0.001	−0.01*	-0.03**
	(-0.49)	(-0.19)	(−1.93)	(-2.01)
Adjusted R-squared	0.47	0.52	0.58	0.68

Table 6: Empirical results—external debt

Note: *** indicates 1%, ** 5%, and * 10/% levels of significance; t-statistics in brackets; FB(-1) = lagged fiscal primary balance, Pub(-1) = lagged public-debt-to-GDP ratio, Ygap = output gap, D1993 and D2001 = dummy variables for 1993 and 2001; J-statistic (GMM eq.1) = 0.99 (0.32); GMM eq. 2 is exactly identified.

Source: authors' own analysis.

Overall, the estimation results do not provide compelling empirical evidence of debt sustainability. The responsiveness of fiscal policy in containing increases in government debt is not strongly supported. However, the finding that increases in debt are positively associated with some improvement in government deficits is consistent with relatively lower national deficits that have been experienced in the past. Nonetheless, the lower deficits could also be attributed to the significant foreign aid support received in the past, which may not necessarily be guaranteed in future.

8 Conclusion and policy implications

Rising public debt in SSA has been a subject of debate regionally and globally. Although Tanzania's public debt has risen rapidly in recent years, it has not attracted much attention in comparison with that of other countries in the region, particularly Kenya and Ethiopia. This paper provides an analysis of public debt and debt sustainability from a macro perspective focusing on external debt. External debt accounts for over 70 per cent of public debt in Tanzania.

Tanzania is among the first countries that benefited from debt relief under the HIPC Initiative and MDRI. By the 1980s, the country was already struggling with debt servicing, and by the 1990s, the

¹⁴ Moreover, it is worth noting that the period of estimation covers the debt relief period.

debt situation had become worse, with public-debt-to-GDP ratios in excess of 100 per cent. Debt relief initiatives reduced the debt burden considerably, thereafter leading to favourable debt indicators and ample space for public borrowing. However, there has been a rapid debt accumulation lately, largely driven by the need to fund infrastructure projects in the country. Although concessional debt still accounts for a higher share, its share has declined to less than 50 per cent while the proportion of commercial and short-term debt has risen. Consequently, the cost of debt servicing is increasing.

Despite the increase, the latest DSA conducted by the government of Tanzania indicates that the risk of public external debt distress is low under the baseline and shock scenarios, albeit sensitive to exchange rate volatility and export shocks. Nonetheless, this DSA is an indication of sensitivity to liquidity risks. Furthermore, the analysis is predicated on strong assumptions of robust GDP growth in the future and a slow-down in government borrowing after completion of the major infrastructure projects under the 'Five-Year Development Plan II'. The empirical evidence of debt sustainability based on estimation results of a standard fiscal reaction function is, however, weak. This implies that, empirically, the responsiveness of fiscal policy in containing increases in government debt is not strongly supported.

Notwithstanding the limitations of debt sustainability frameworks, the assessment underscores the sound macroeconomic and fiscal management that has been associated with robust growth in recent period. However, the main challenge lies in ensuring that debt remains sustainable for the foreseeable future, given the need to scale up development projects to address infrastructure gaps amid dwindling donor financing, changing debt structure, and a fragile global economy that has been hit hard by the COVID-19 pandemic. There is no guarantee that the strong assumptions that underpin the DSA will remain constant. A rapid external debt accumulation, particularly of commercial debt, could expose the country to external vulnerabilities and liquidity pressures relating to debt obligations. Given the rising cost of debt servicing, reliance on commercial debt could be used carefully and selectively, so as not to constrain the scope for enhancing socioeconomic progress in other areas of development. It is vital to leverage on concessional borrowing, coupled with enhanced efficiency of public investment and debt management. There is a need to build capacity for managing an increasingly market-based debt profile and the associated risks, particularly as Tanzania continues to gravitate towards non-concessional sources of credit.

Strategies aimed at increasing domestic resource mobilization are critical. Besides tax reform initiatives, these should include modalities for public-private partnership. The revamping of foreign exchange earnings from the export of goods and services will, besides enhancing the country's ability to service debt, boost economic growth.

References

- Atingi, M. (2019). Public Debt Accumulation in Africa: A Looming Debt' Paper presented at the Plenary Session during the AERC (African Economic Research Consortium) Biannual Research Workshop, Cape Town, 2–5 June 2019.
- Bank of Tanzania (2020). 'Monetary Policy Statement: Mid-Term Review 2019/20'. Dar es Salaam: Bank of Tanzania.
- Bank of Tanzania (various dates). Annual Reports, Monthly Economic Reviews, and Quarterly Economic Bulletins. Dar es Salaam: Bank of Tanzania.

- Bartoletto, S., B. Chiarini, and E. Marzano (2013). 'Is the Italian Public Debt Really Unsustainable? An Historical Comparison (1861–2010)'. CESifo Working Paper 4185. Munich: Centre for Economic Studies and Ifo Institute (CESifo).
- Bassanetti, A., C. Cottarelli, and A.F. Presbitero (2019). Lost and Found: Market Access and Public Debt Dynamics'. Oxford Economic Papers, 71(2): 445–71. https://doi.org/10.1093/oep/gpy040
- Beqiraj, E., S. Fedeli, and F. Forte (2018). 'Public Debt Sustainability: An Empirical Study on OECD Countries'. *Journal of Macroeconomics*, 58: 238–48. https://doi.org/10.1016/j.jmacro.2018.10.002
- Bohn, H. (1998). 'The Behavior of US Public Debt and Deficits'. The Quarterly Journal of Economics, 113(3): 949–63. https://doi.org/10.1162/003355398555793
- Bohn, H. (2007). 'Are Stationary and Cointegration Restrictions Really Necessary for the Intertemporal Budget Constraint?' *Journal of Monetary Economics*, 54(7): 1837–47. https://doi.org/10.1016/j.jmoneco.2006.12.012
- Bökemeier, B., and A. Stoian (2018). 'Debt Sustainability Issues in Central and East European Countries'. *Eastern European Economics*, 56(5): 438–70. https://doi.org/10.1080/00128775.2018.1496456
- Calderisi, R. (2015). The Trouble with Africa: Why Foreign Aid Isn't Working. London: Yale University Press.
- Cassimon, D., V.B. Campenhout, M. Ferry, and M. Raffinot (2015). Africa: Out of Debt, into Fiscal Space? Dynamic Fiscal Impact of the Debt Relief Initiatives on African Heavily Indebted Poor Countries (HIPCs)'. International Economics, 144: 29–52. https://doi.org/10.1016/j.inteco.2015.04.007
- Chatterjee, S., J. Gibson, and F. Rioja (2017). 'Optimal Public Debt Redux'. Journal of Economic Dynamics & Control, 83: 162–74. https://doi.org/10.1016/j.jedc.2017.08.005
- Collignon, S. (2012). 'Fiscal Policy Rules and the Sustainability of Public Debt in Europe'. International Economic Review, 53(2): 539-67. https://doi.org/10.1111/j.1468-2354.2012.00691.x
- Coulibaly, B.S., D. Gandhi, and L.W. Senbet (2019). 'Is Sub-Saharan Africa Facing Another Systemic Sovereign Debt Crisis?' Africa Growth Initiative Policy Brief, April. Washington, DC: Brookings Institution.
- ICTD (International Centre for Tax and Development) and UNU-WIDER (2019). 'GRD—Government Revenue Dataset'. Available at: https://www.wider.unu.edu/project/ government-revenue-dataset (accessed October 2019).
- IMF (2001). 'IMF and World Bank Support for U\$3 Billion in Debt Service Relief for Tanzania under Enhanced HIPC Initiative'. Press release, 27 November. Washington, DC: IMF. Available at: https://www.imf.org/en/News/Articles/2015/09/14/01/49/pr0148 (accessed November 2019).
- IMF (2014). 'Revised Guidelines for Public Debt Management'. IMF Policy Paper 14/002. Washington, DC: IMF. https://doi.org/10.5089/9781498343602.007
- IMF (2018a). Regional Economic Outlook for Sub-Saharan Africa: Domestic Revenue Mobilization and Private Investment, April 2018. Washington, DC: IMF.
- IMF (2018b). 'United Republic of Tanzania: Seventh Review under the Policy Support Instrument—Press Release; Staff Report'. IMF Country Report 18/11, January. Washington, DC: IMF. https://doi.org/10.5089/9781484337981.002
- IMF (2019). 'Heavily Indebted Poor Countries (HIPC) Initiative and Multilateral Debt Relief Initiative (MDRI)—Statistical Update'. Policy Paper, August. Washington, DC: IMF.
- IMF (2020). 'Factsheet: Joint World Bank-IMF Debt Sustainability Framework for Low-Income Countries'. Available at: https://www.imf.org/en/About/Factsheets/Sheets/2016/08/01/16/39/Debt-Sustainability-Framework-for-Low-Income-Countries (accessed March 2020).
- Jayme Jr, F.G. (2001). 'External Debt Sustainability: Empirical Evidence in Brazil'. Texto para discussão 154. Belo Horizonte: Cedeplar—Centro de Desenvolvimento e Planejamento Regional da UFMG (Universidade Federal de Minas Gerais).

- Kaiser, J., I. Knoke, and H. Kowsky (2009). 'Towards a Renewed Debt Crisis? Risk Profiles of the Poorest Countries in the light of the Global Economic Slowdown.' Occasional Paper 44. Berlin: Friedrich-Ebert-Stiftung.
- Karam, P.D., and D. Hostland (2006). 'Specification of a Stochastic Simulation Model for Assessing Debt Sustainability in Emerging Market Economies. Working Paper 06/028. Washington, DC: IMF. https://doi.org/10.5089/9781451865288.001
- Kayandabila, Y., and M.K. Manyama (2013). 'Fiscal Policy and Debt Dynamic: Evidence from Tanzania'. *African Journal of Economic Review*, 1(2): 30–46.
- Llorca, M. (2017). 'External Debt Sustainability and Vulnerabilities: Evidence from a Panel of 24 Asian Countries and Prospective Analysis'. Working Paper 692. Tokyo: Asian Development Bank Institute.
- Mahmood, T., S.A. Rauf, and H.K. Ahmad (2009). 'Public and External Debt Sustainability in Pakistan (1970s-2000s)'. *Pakistan Economic and Social Review*, 47(2): 243-67.
- Makin, A.J., and J. Pearce (2014). 'How Sustainable Is Sub-National Public Debt in Australia?' *Economic Analysis and Policy*, 44(4): 364–75. https://doi.org/10.1016/j.eap.2014.10.001
- Melina, G., S.S. Yang, and L. Zanna (2016). 'Debt Sustainability, Public Investment, and Natural Resources in Developing Countries: The DIGNAR Model'. *Economic Modelling*, 52: 630–49. https://doi.org/10.1016/j.econmod.2015.10.007
- Ministry of Finance and Planning (2019). Tanzania National Debt Sustainability Analysis (December, 2019). Dar es Salaam: Government of Tanzania.
- Mkapa, B.W. (2019). My Life, My Purpose: A Tanzanian President Remembers. Dar es Salaam: Mkuki na Nyota Publishers.
- Mollel, L. (2009). 'Impact of Debt Relief on Fiscal Allocation to Social Priority Sectors and Response of Social Indicators in the HIPCs: A Case Study of Tanzania'. Master's Dissertation, University of Cape Town, South Africa.
- Mustapha, S., and A. Prizzon (2015). *Debt Sustainability and Debt Management in Developing Countries*. London: Overseas Development Institute.
- Naraidoo, R., and L. Raputsoane (2015). 'Debt Sustainability and Financial Crises in South Africa'. *Emerging* Markets Finance and Trade, 51: 224–33. https://doi.org/10.1080/1540496X.2015.1011534
- Ncube, M., and Z. Brixiová (2015). 'Public Debt Sustainability in Africa: Building Resilience and Challenges Ahead'. *Development Policy Review*, 33(5): 555–80. https://doi.org/10.1111/dpr.12126
- Ndulu, B. (2018). 'The Conduct of Monetary Policy in East Africa in a Changing Environment'. Memorial Lecture in Honour of Professor Francis Mwega, University of Nairobi.
- Ndulu, B.J., and S.A. O'Connell (2019). 'Africa's Development Debts'. Paper presented at the Plenary Session of the African Economic Research Consortium (AERC) Biannual Research Workshop, Cape Town, 2 June.
- Nkhata, S. (2009). 'Analysis of the Main Risks to Public Debt Sustainability in the MEFMI Post-HIPC Countries: The Case Study of Malawi, Tanzania and Uganda'. Paper submitted to Macroeconomic and Financial Management Institute of Eastern and Southern Africa (MEFMI) in partial fulfilment of the requirements of the MEFMI Fellows Development Programme.
- Nord, R., Y. Sobolev, D. Dunn, A. Hajdenberg, N. Hobdari, S. Maziad, and S. Roudet (2009). *Tanzania: The Story of an African Transition*. Washington, DC: IMF.
- Paret, A.C. (2017). 'Debt Sustainability in Emerging Market Countries: Some Policy Guidelines from a Fan-Chart Approach'. *Economic Modelling*, 63: 26–45. https://doi.org/10.1016/j.econmod.2017.01.010
- Pinto, B. (2018). 'The Distressing Debt Sustainability Framework of the IMF and World Bank' Blog article, 3 July. Washington, DC: Brookings Institution.
- Pradhan, K. (2019). 'Analytical Framework for Fiscal Sustainability: A Review'. Review of Development and Change, 24(1): 100–22. https://doi.org/10.1177/0972266119845951

- Tiwari, A.K. (2012). 'Debt Sustainability in India: Empirical Evidence Estimating Time-Varying Parameters'. *Economics Bulletin*, 32(2): 1133–41.
- Tóth, C.G. 2011. 'Debt Dynamics and Sustainability'. *Hungarian Statistical Review*, Special Number 16 (translated version).
- Trehan, B., and C. Walsh (1988). 'Common Trends, the Government Budget Constraint, and Revenue Smoothing.' *Journal of Economic Dynamics and Control*, 12: 425–24. https://doi.org/10.1016/0165-1889(88)90048-6
- Stiglitz, J.E. (2016). 'Sovereign Debt: Notes on Theoretical Frameworks and Policy Analyses'. Working Paper 301. New York: Initiative for Policy Dialogue, Columbia University.
- Sundell, L., and M. Lemdal (2011). 'Debt Overhang and the Effects on Developing and Developed Economies' Stockholm: Department of Economics, Stockholm School of Economics.
- World Bank (2019). "The World Bank in Tanzania: Overview'. Dar es Salaam: World Bank. Available at: http://www.worldbank.org/en/country/tanzania/overview (accessed October 2019).
- World Bank (2020a). 'International Debt Statistics'. Available at: https://datatopics.worldbank.org/debt/ids (accessed February 2020).
- World Bank (2020b). 'World Development Indicators'. Available at: https://databank.worldbank.org/source/world-development-indicators (accessed February 2020).